Author's response to reviews

Title: Comparison of emergency department and hospital admissions data for air pollution time series studies

Authors:

Andrea Winquist (awinqui@emory.edu)
Mitchel Klein (mklein@emory.edu)
Paige E. Tolbert (ptolber@sph.emory.edu)
W. Dana Flanders (wflande@sph.emory.edu)
Jeremy Hess (jhess@emory.edu)
Stefanie E. Sarnat (sebelt@emory.edu)

Version: 4 Date: 31 August 2012

Author's response to reviews: see over
Dear Editors,

Thank you for considering our manuscript revision. We would like to submit the accompanying manuscript for your consideration. It has been further revised to address formatting issues. The following changes have been made and all changes are indicated in the manuscript using 'tracked changes'.

- Removed line numbering.
- Corrected formatting of title page, abstract, and abbreviation list, as requested.
- Corrected title for “Author’s contributions” section.
- Added horizontal lines to tables and removed shading.
- Changed use of capitalization in headings throughout the manuscript to conform to journal style (capitalized only first word of headings)
- Corrected one reference in which ‘et al’ was used before 30 authors had been listed.
- Corrected three journal abbreviations in references (previously not abbreviated).

Thank you for your consideration of our revised manuscript.

Sincerely,

Andrea Winquist
Assistant Research Professor
Rollins School of Public Health, Department of Environmental Health